Author's response to reviews

Title: Idiopathic toe walking and sensory processing dysfunction

Authors:

Cylie M Williams Ms (cylie.williams@southernhealth.org.au)
Paul Tinley Assoc Prof (ptinley@csu.edu.au)
Michael Curtin Dr (mcurtin@csu.edu.au)

Version: 4 Date: 10 August 2010

Author's response to reviews:

11/08/10

The Independent Editorial Production Team
Tel: +44 (0) 20 3192 2014
Email: independentprepublication@biomedcentral.com
Web: http://www.jfootankleres.com/

Re: “Idiopathic toe walking and sensory processing dysfunction”
Please find the following revisions:

Title: please ensure that the title is identical in the manuscript and the submission system
Title has been modified within submission system

Authors’ contributions: Please place the Authors’ contributions section towards the end of the manuscript text, directly preceding the Acknowledgments.
Contributions as been added to manuscript preceding the acknowledgments

References: please provide functioning links to the websites in references 15 and 50: Functioning links have been added to references 15 and 50

References: References should be cited in the text using consecutive numbers in square brackets. All numbers have been checked and included with brackets

Minor revisions

Qualifications: Please remove the authors' qualifications (e.g. PhD / MD) or job titles from the manuscript and submissions system. Qualifications and Job titles have been removed.

Affiliations: Please remember to include the country in the affiliation details. Country has been included in all affiliations

Tables: Please remove the visible vertical lines from the tables and please note that the table title should only be one sentence long. Vertical lines have been
removed from tables

Blank pages: Please remove the blank pages on page 18 and 24. Blank pages have been removed.

Kind Regards,
Cylie Williams